Math 132 Winter 2004 R. Schultz

UPDATED GENERAL INFORMATION — FEBRUARY 4, 2004 Here are the results for the first examination (P = total points possible): GRADE CUTOFF POINTS P: 100 A: 80 B: 59 C: 43 D: 25 The median score was in the mid sixties, and the average was slightly lower. Requests for reconsideration must be submitted in writing no later than one week after the return of the examination papers. Any explanations of reasons should be brief and deal with the specific mathematical points involved. Solutions for the examination problems will be posted in the file exam1a.pdf in the course directory. Solutions for some of the additional exercises in the third set will be posted in the file solutions3.pdf in the course directory.

Here is the fourth homework assignment, which is due in class on Friday, February 13, 2004. “F&B” refers to the course text by Fraleigh and Beauregard.

Exercises 4 and 22 beginning on p. 358 of F&B. Additional Exercise 1 for Section II.3 Additional Exercise 1 for Section III.A Exercise 10 beginning on p. 394 of F&B. Exercise 2 beginning on p. 406 of F&B. Additional Exercises 1 and 2a for Section III.1 Exercise 26 beginning on p. 368 of F&B. Additional Exercise 1b for Section III.2 Exercises 6, 16 and 22 beginning on p. 463 of F&B. Additional Exercise 1 for Section IV.1